Author's response to reviews

Title: A Thyroid Tumor extending to the Parapharyngeal Space

Authors:

Fikret Cetik (fikretcetik@yahoo.com.tr)
Demet Yazici (demetyazici@yahoo.com)
Aysun Uguz (aysunuz@yahoo.com)

Version: 3 Date: 24 January 2006

Author's response to reviews:

We have changed the abstract of the article according to the BMC format and added the written consent of the patient at the end of the article.